\begin{table}%T1
\caption{\label{tbl-1}O--C data for T~Antliae.}
\begin{tabular}{cccccll}
\hline\hline
\noalign{\smallskip}
{\rm HJD}$_{\max}$ &{\rm Cycles}, $E$ &$C$ &${\rm O}{-}{\rm C}$ &{\rm Weight} 
&{\rm Data Type} &{\rm Source} \\
& & &{\rm(days)} & &{\rm(observations)} & \\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
2~415~165.745 &-3553 &2~415~164.826 &+0.924 &1.0 &{\rm pg(18)} 
&{\rm This paper} \\
2~416~079.895 &-3398 &2~416~079.024 &+0.871 &1.0 &{\rm pg(29)} 
&{\rm This paper} \\
2~416~563.482 &-3316 &2~416~562.664 &+0.818 &1.0 &{\rm pg(19)} 
&{\rm This paper} \\
2~416~887.751 &-3261 &2~416~887.057 &+0.694 &1.0 &{\rm pg(18)} 
&{\rm This paper} \\
2~417~418.653 &-3171 &2~417~417.882 &+0.771 &1.0 &{\rm pg(23)} 
&{\rm This paper} \\
2~418~031.991 &-3067 &2~418~031.279 &+0.712 &1.0 &{\rm pg(6)} 
&{\rm This paper} \\
2~418~409.325 &-3003 &2~418~408.755 &+0.570 &1.0 &{\rm pg(22)} 
&{\rm This paper} \\
2~418~751.489 &-2945 &2~418~750.842 &+0.647 &1.0 &{\rm pg(17)} 
&{\rm This paper} \\
2~419~435.610 &-2829 &2~419~435.016 &+0.594 &1.0 &{\rm pg(25)} 
&{\rm This paper} \\
2~420~096.127 &-2717 &2~420~095.598 &+0.529 &1.0 &{\rm pg(30)} 
&{\rm This paper} \\
2~420~638.730 &-2625 &2~420~638.219 &+0.511 &1.0 &{\rm pg(25)} 
&{\rm This paper} \\
2~421~039.753 &-2557 &2~421~039.286 &+0.467 &1.0 &{\rm pg(31)} 
&{\rm This paper} \\
2~421~370.151 &-2501 &2~421~369.577 &+0.574 &1.0 &{\rm pg(21)} 
&{\rm This paper} \\
2~421~741.631 &-2438 &2~421~741.155 &+0.476 &1.0 &{\rm pg(26)} 
&{\rm This paper} \\
2~422~071.833 &-2382 &2~422~071.446 &+0.387 &1.0 &{\rm pg(19)} 
&{\rm This paper} \\
2~422~449.308 &-2318 &2~422~448.921 &+0.387 &1.0 &{\rm pg(18)} 
&{\rm This paper} \\
2~423~098.073 &-2208 &2~423~097.707 &+0.366 &1.0 &{\rm pg(10)} 
&{\rm This paper} \\
2~423~723.252 &-2102 &2~423~722.901 &+0.351 &1.0 &{\rm pg(29)} 
&{\rm This paper} \\
2~424~472.287 &-1975 &2~424~471.953 &+0.334 &1.0 &{\rm pg(27)} 
&{\rm This paper} \\
2~425~333.562 &-1829 &2~425~333.069 &+0.493 &1.0 &{\rm pg(24)} 
&{\rm This paper} \\
2~427~598.158 &-1445 &2~427~597.921 &+0.237 &1.0 &{\rm pg(16)} 
&{\rm This paper} \\
2~428~276.479 &-1330 &2~428~276.198 &+0.281 &1.0 &{\rm pg(22)} 
&{\rm This paper} \\
2~428~582.994 &-1278 &2~428~582.896 &+0.098 &1.0 &{\rm pg(11)} 
&{\rm This paper} \\
2~429~178.759 &-1177 &2~429~178.600 &+0.159 &1.0 &{\rm pg(30)} 
&{\rm This paper} \\
2~430~057.629 &-1028 &2~430~057.410 &+0.219 &1.0 &{\rm pg(26)} 
&{\rm This paper} \\
2~430~629.675 &-931 &2~430~629.521 &+0.154 &1.0 &{\rm pg(25)} 
&{\rm This paper} \\
2~431~260.758 &-824 &2~431~260.612 &+0.146 &1.0 &{\rm pg(25)} 
&{\rm This paper} \\
2~431~608.806 &-765 &2~431~608.597 &+0.209 &1.0 &{\rm pg(21)} 
&{\rm This paper} \\
2~431~927.259 &-711 &2~431~927.092 &+0.167 &1.0 &{\rm pg(17)} 
&{\rm This paper} \\
2~432~310.670 &-646 &2~432~310.466 &+0.204 &1.0 &{\rm pg(10)} 
&{\rm This paper} \\
2~432~705.716 &-579 &2~432~705.635 &+0.081 &1.0 &{\rm pg(21)} 
&{\rm This paper} \\
2~433~006.564 &-528 &2~433~006.436 &+0.128 &1.0 &{\rm pg(13)} 
&{\rm This paper} \\
2~433~507.869 &-443 &2~433~507.771 &+0.098 &1.0 &{\rm pg(9)} 
&{\rm This paper} \\
2~434~221.527 &-322 &2~434~221.435 &+0.092 &1.0 &{\rm pg(17)} 
&{\rm This paper} \\
2~434~887.978 &-209 &2~434~887.915 &+0.063 &3.0 &{\rm pe(7)} 
&{\rm Walraven et al. (\cite{we58})} \\
2~435~566.277 &-94 &2~435~566.191 &+0.086 &3.0 &{\rm pe(44)}
&{\rm Walraven et al. (\cite{we58})} \\
2~437~005.435 &+150 &2~437~005.316 &+0.119 &3.0 &{\rm pe(11)} 
&{\rm Irwin (\cite{ir61})} \\
2~437~648.231 &+259 &2~437~648.204 &+0.027 &3.0 &{\rm pe(5)} 
&{\rm Mitchell et al. (\cite{mi64})} \\
2~440~780.086 &+790 &2~440~780.070 &+0.016 &3.0 &{\rm pe(31)} 
&{\rm Pel (\cite{pe76})}$^{{a}}$ \\
2~444~018.190 &+1339 &2~444~018.101 &+0.089 &2.0 &{\rm pe(5)} 
&{\rm Harris (\cite{ha80})} \\
2~444~200.988 &+1370 &2~444~200.941 &+0.047 &2.0 &{\rm pe(8)} 
&{\rm Caldwell et al. (\cite{ca01})} \\
2~444~666.950 &+1449 &2~444~666.887 &+0.063 &3.0 &{\rm pe(19)} 
&{\rm Eggen (\cite{eg85})} \\
2~447~952.239 &+2006 &2~447~952.102 &+0.137 &3.0 &{\rm IDD(42)} 
&{\rm ESA (\cite{esa97})} \\
2~448~123.280 &+2035 &2~448~123.146 &+0.134 &3.0 &{\rm IDD(37)} 
&{\rm ESA (\cite{esa97})} \\
2~448~424.078 &+2086 &2~448~423.947 &+0.131 &3.0 &{\rm IDD(77)} 
&{\rm ESA (\cite{esa97})} \\
2~448~689.538 &+2131 &2~448~689.359 &+0.179 &3.0 &{\rm IDD(49)} 
&{\rm ESA (\cite{esa97})} \\
2~449~019.831 &+2187 &2~449~019.650 &+0.181 &3.0 &{\rm IDD(23)} 
&{\rm ESA (\cite{esa97})} \\
2~451~644.531 &+2632 &2~451~644.284 &+0.247 &3.0 &{\rm pe(24)} 
&{\rm Berdnikov \& Caldwell (\cite{bc01})} \\
2~451~963.024 &+2686 &2~451~962.778 &+0.246 &3.0 &{\rm pe(26)} 
&{\rm Berdnikov \& Turner (\cite{bt01})} \\
2~452~358.255 &+2753 &2~452~357.948 &+0.307 &3.0 &{\rm pe(67)} 
&{\rm Berdnikov \& Turner (\cite{bt02})} \\
2~452~647.252  &+2802 &2~452~646.952 &+0.300 &3.0 &{\rm pe(39)} 
&{\rm Berdnikov \& Turner (\cite{bt03})} \\
\hline
\end{tabular}
\smallskip
$^{{a}}$ Photometry used as reference standard.
\end{table}